Author's response to reviews

Title: Cardiovascular disease prevention in rural Nigeria in the context of a community based health insurance scheme: QUality Improvement Cardiovascular care Kwara-I (QUICK-I)

Authors:

Marleen Hendriks (m.hendriks@amc-cpcd.org)
Lizzy Brewster (L.M.Brewster@amc.uva.nl)
Ferdinand Wit (f.wit@amc-cpcd.org)
Oladimeji Akeem Bolarinwa (drdeji@yahoo.com)
Aina Olufemi Odusola (A.O.Odusola@amc.uva.nl)
William Redekop (redekop@bmg.eur.nl)
Navin Bindraban (n.r.bindraban@amc.uva.nl)
Albert Vollaard (a.vollaard@pharmaccess.org)
Shade Alli (sadealling@yahoo.com)
Peju Adenusi (dradenusi@hygeia.com)
Kayode Agbede (agbededejk@yahoo.com)
Tanimola Akande (akandetm@yahoo.com)
Joep Lange (j.lange@amc.uva.nl)
Constance Schultsz (schultsz@gmail.com)

Version: 2 Date: 1 March 2011

Author's response to reviews:

Dear Editor, dear Mrs Pafitis,

In response to your email dated 17-2-2011 (MS: 1281588731518325) we have revised the 2 QUICK manuscripts such that the sister manuscript is included in the reference list of each paper and that they are cross cited.

On behalf of all authors,

Yours sincerely,

Aina Olufemi Odusola and Marleen Hendriks